Author's response to reviews

Title: Impact of Socioeconomic Status and Medical Conditions on Subjective Health and Healthcare Utilization among Older Adults in Ghana

Authors:

Bashiru I. I Saeed (sirbash156@gmail.com)
Alfred Edwin Yawson (aeyawson@yahoo.com)
Samuel Blay Nguah (sbnquah@gmail.com)
Edmund Ayesu (eayesu76@gmail.com)

Version: 2 Date: 28 October 2014

Author's response to reviews: see over
October 26, 2014.

The Editor
BMCPublic Health

Dear Sir/Madam,

Kindly find attached the manuscript ‘Impact of socioeconomic status and medical conditions on subjective health and healthcare utilization among ageing Ghanaians’.

This study attempts to examine the impact of socioeconomic and medical conditions in health and healthcare utilization among ageing Ghanaians. We tried as much as possible to follow the Journal's Instructions for authors; we are convinced this work falls within the scope of this journal. We hope that you will consider this manuscript for publication in your reputable journal. Kindly direct all correspondence with regard this manuscript to:

Bashiru I.I. Saeed
Tel: +233 (0)208204092
Email address: sirbash156@gmail.com

We wait for your response and the comments of reviewers.

Yours sincerely
Bashiru I.I. Saeed

(Corresponding Author)
Corresponding Author:
Bashiru I.I. Saeed\textsuperscript{1,2}
\textit{Tel:++86 15606109182}
Email address: \textit{sirbash156@gmail.com}

Author name(s), Email and affiliation(s)
1. Bashiru I.I. Saeed\textsuperscript{1,2}
\textit{sirbash156@gmail.com}
2. Alfred Yawson\textsuperscript{3}
\textit{aeyawson@yahoo.com}

3. Samuel Blay Nguah\textsuperscript{4}
\textit{sbnguah@gmail.com}
4. Edmund Ayesu\textsuperscript{5}
\textit{eayesu76@gmail.com}

Affiliate Institution:
\textsuperscript{1} School of Finance and Economics, Jiangsu University, Zhenjiang, China
\textsuperscript{2} Mathematics and Statistics Department, Kumasi Polytechnic, Kumasi, Ghana

\textsuperscript{3}Department of Community Health, University of Ghana Medical School, College of Health Sciences, Korle-Bu, Accra, Ghana

\textsuperscript{4} Paediatric Department, Komfo Anokye Teaching Hospital, Kumasi, Ghana

\textsuperscript{5} Graduate School and Research Centre, Kumasi Polytechnic, Kumasi, Ghana
Postal Address:

O A Road, Kumasi Polytechnic,

*Faculty of Applied Sciences*

P O BOX 854

Kumasi, Ghana